Name of Journal: World Journal of Methodology

Manuscript NO: 34409

Manuscript Type: Original Article

We would like to thank the reviewers for the nice comments on the paper.

The editing team suggestions have all been addressed.

Thank you for the helpful suggestions.

Best regards,

Francesco Doglietto, MD, PhD, Assistant Professor

Neurosurgery, Department of Medical and Surgical Specialties, Radiological Sciences and Public Health, University of Brescia, Piazzale Spedali Civili, 1, 25123 Brescia, Italy. francesco.doglietto@unibs.it





To: Professor Yicheng Ni, MD,PhD Editor-in-Chief, World Journal of Methodology

Brescia, September 1, 2017

Dear Professor Yicheng Ni,

I am submitting to World Journal of Methodology the revised basic study entitled "Quantification and comparison of cranial approaches in the anatomy laboratory: a neuronavigation-based, research method".

I have modified the title of the manuscript according to your suggestion and added the CT scan manufacturer data.

I would like to thank you for your suggestions and support. Best regards,

Francesco Doglietto, MD, PhD

Assistant Professor

Neurosurgery, Department of Surgical Specialties, Radiological Sciences, and Public Health University of Brescia

Brescia, Italy